Author's response to reviews

Title: Soluble Fas/Fas Ligand might serve as a diagnostic tool for gastric adenocarcinoma

Authors:

- Samaneh Boroumand-Noughabi (s_n_bromand@yahoo.com)
- Hamid Reza Sima (simahr@mums.ac.ir)
- Kamran Ghaffarzadehgan (kghafarzadegan@hotmail.com)
- Mostafa Jafarzadeh (mostafa.jafarzadeh@gmail.com)
- Hamid Reza Raziee (razieehr@mums.ac.ir)
- Hanieh Hosseinnezhad (hosseinnezhad.hanieh@gmail.com)
- Omeed Moaven (omeedmoaven@gmail.com)
- Mohammad Taghi Rajabi-Mashhadi (rajabimt@mums.ac.ir)
- Amir Abbas Azarian (azarianaa1@mums.ac.ir)
- Mojtaba Mashhadinejad (m.mashhadinejad@gmail.com)
- Jalil Tavakkol-Afshari (tavakolaj@mums.ac.ir)

Version: 3 Date: 21 May 2010

Author's response to reviews: see over
May 21, 2010

Dear Editor,

Per your request of performing changes to the format of the manuscript # 2849049063040744 according to the journal style, I am resubmitting this manuscript including the changes listed below, based on the provided comments:

- Title is identical in the manuscript and the submission system
- Table 1 was submitted as an additional file as it was large and laid out as landscape. In the resubmission, table one is split into two tables and put at the end of the manuscript. Consequently no additional files are submitted.
- Authors' qualifications (e.g. PhD / MD) or job titles are removed from the manuscript and submissions system

Sincerely yours,

Hamid Reza Sima, M.D
Director, Gastroenterology and Hepatology fellowship
Assistant professor of Medicine, 
Division of GI and Liver Disease, Department of Internal Medicine
Imam-Reza Hospital
Mashhad University of Medical Sciences,
Mashhad, Iran
Tel: +98 511 844 6156 
Fax: +98 511 859 8818
24, 10th Pasteur St.
Mashhad 91837, Iran
Email: simahr@mums.ac.ir